Author's response to reviews

Title: Magnetic resonance imaging in size assessment of invasive breast carcinoma with an extensive intraductal component.

Authors:

Arjan P Schouten van der Velden (aschoutenvandervelden@hotmail.com)
Carla Boetes (c.boetes@mumc.nl)
Peter Bult (p.bult@pathol.umcn.nl)
Theo Wobbes (t.wobbes@chir.umcn.nl)

Version: 2 Date: 3 March 2009

Author's response to reviews: see over
Magnetic resonance imaging in size assessment of invasive breast carcinoma with an extensive intraductal component.

Arjan P Schouten van der Velden, Carla Boetes, Peter Bult and Theo Wobbes

Editor’s comments.

The manuscript (e.g. names on title page, references) has been adjusted to meet journal style.

The conflict of interest statement has been changed in ‘Competing Interests statement’.

Reviewer #1 [Zuiani]

The methods section has been changed and some further technical details about the MRI imaging protocol are included.

The measuring of tumour extent on mammography is based on ‘in vivo’ mammographic exams. E.g., it is not based on findings from the histopathological examination during which breast tissue is cross-sectioned and X-rays are performed.

Figure 2 has been enclosed.

Reviewer #2 [Chen]

No further comments.